Author’s response to reviews

Title: Prevalence of visual impairment, cataract surgery and awareness of cataract and glaucoma in Bhaktapur district of Nepal: The Bhaktapur Glaucoma Study

Authors:

Suman S. Thapa (ssthapap@tilganga.com.np)
Rosa V Berg (rosavandenberg@hotmail.com)
Shankar Khanal (shankar_cds@yahoo.com)
Indira Paudyal (indurup@gmail.com)
Pooja Pandey (ppandey@hki.org)
Nhukesh Maharjan (nhukesh@tilganga.com.np)
Shankha Twayna (eyebank@tilganga.com.np)
Govinda Paudyal (govinda@tilganga.com.np)
Reeta Gurung (reeta@tilganga.com.np)
Sanduk Ruit (sruit@tilganga.com.np)
Ger H.M.B. van Rens (rens@vumc.nl)

Version: 6 Date: 5 January 2011

Author’s response to reviews:

1. Author list: the list of authors in the manuscript has been rectified.

2. Abstract: Abstract present in the manuscript matches exactly with the submission system.

3. Authors’ contributions: All authors have read and approved the final manuscript which is included in the manuscript.

4. Table 8: Has been reformatted.

5. Highlighting/tracking: we have removed all highlighting and tracking from the manuscript.

6. References: the reference list has been corrected as per your advice.